Health state utilities associated with adult attention-deficit/hyperactivity disorder

Louis S Matza¹
Mary K Devine¹
Virginia Sutton Haynes²
Evan W Davies³
Jacqueline M Kostelec¹
Foula Televantou⁴
Jessica B Jordan¹

Outcomes Research, Evidera, Bethesda, MD, USA; ²Eli Lilly and Company, Indianapolis, IN, USA; ³Outcomes Research, Evidera, London, UK; ⁴Eli Lilly and Company, Windlesham, Surrey, UK **Objectives:** With growing awareness of the importance of adult attention-deficit/hyperactivity disorder (ADHD) treatment, cost-effectiveness analyses, including utilities, are needed to compare the value of treatment options. Although utilities have been reported for childhood ADHD, little is known about utilities representing adult ADHD. Therefore, the purpose of this study was to estimate utilities associated with adult ADHD.

Methods: Health-state descriptions of adult ADHD were drafted based on literature review, interviews with four clinicians, and clinical trial data. Health states were revised based on a pilot study with 26 participants. Final health states were rated in time trade-off interviews with general population respondents in London and Edinburgh, UK.

Results: A total of 158 participants completed interviews (mean age =47.0 years; 49.4% female; Edinburgh =80 participants). Mean (standard deviation [SD]) utilities were 0.82 (0.17), 0.68 (0.28), and 0.67 (0.28) for health states describing treatment responders (health state A), nonresponders (health state B), and untreated patients (health state C), respectively. Most participants rated health state A as preferable to B (n=92; 58.2%) and C (n=97; 61.4%). The majority rated B and C as equal (n=125; 79.1%). Paired Student's *t*-tests found that A had a significantly greater mean utility than B (t=10.0; t<0.0001) and C (t=10.2; t<0.0001).

Conclusion: The current study provides utilities that may be used in cost—utility models of treatment for adult ADHD. Results reflected clear differences between health states representing treatment responders and nonresponders/untreated patients. Current utilities were comparable to those previously reported for childhood ADHD.

Keywords: adult ADHD, utility, time trade-off

Introduction

A growing body of research indicates that symptoms of attention-deficit/hyperactivity disorder (ADHD) often persist into adulthood.¹⁻⁴ With increasing awareness of the need to treat adult ADHD, cost-effectiveness analyses are needed to demonstrate and compare the value of new and existing treatment options. A cost–utility model is a type of cost-effectiveness analysis that requires utilities, which are values on a scale with anchors of 1 (full health) and 0 (dead) to represent the strength of preferences for various health states.⁵

Although utilities have been reported for childhood ADHD, including values associated with various levels of treatment response, ⁶⁻⁹ little is known about utilities associated with adult ADHD. Two recent reviews reported no published utility values representing ADHD in adults. ^{10,11} Literature searches conducted for the current study also found no published work on adult ADHD utilities, but one conference presentation was located that reported adult ADHD utilities derived from SF-36® (Medical Outcomes Trust, Hanover, NH, USA) responses via the SF-6D scoring algorithm. ¹² However, the six items from the SF-6D do not assess attention or other constructs that are specifically

Correspondence: Louis S Matza Evidera, 7101 Wisconsin Avenue, Suite 600, Bethesda, MD 20814, USA Tel +1 301 664 7263 Fax +1 301 654 9864 Email louis.matza@evidera.com relevant to ADHD, ¹³ which suggests that it may have limited content validity for this particular condition. Therefore, the purpose of this study was to estimate utilities associated with adult ADHD using a direct utility elicitation method. Time trade-off (TTO) methods were used to obtain utilities for health states representing treatment responders, nonresponders, and untreated patients.

Methods

Overview of study design

Adult ADHD health-state descriptions (often called vignettes) were drafted based on literature review, clinical trial results, and input from clinicians. These health states were refined based on additional clinician interviews and a pilot study conducted with general population respondents in London, UK. Utilities were then derived in a TTO valuation study with general population participants in Edinburgh and London, UK.

The vignette-based approach was used because it is well-suited for obtaining utility estimates for specific conditions. Other commonly used utility assessment approaches may have limited applicability to ADHD. For example, generic preference-based instruments such as the EQ-5D (EuroQol Group, Rotterdam, the Netherlands) do not have dimensions assessing ADHD symptoms such as inattention or impulsivity. Furthermore, mapping an ADHD-specific questionnaire to a generic preference-based measure would be severely limited by the lack of overlapping content between these two types of instruments. In addition, no ADHD-specific preference-based measure is available for obtaining utilities directly from this population.

Health states

Three health states describing adult ADHD were rated in the TTO interviews: treatment responder (A), nonresponder (B), and untreated (C). All health states include a parallel series of statements in four areas: diagnosis of adult ADHD, treatment, symptoms, and impact on functioning. The responder and nonresponder health states both included the same statement indicating that treatment is being received. This treatment statement was nonspecific so that the utility values could be used to represent either pharmaceutical or nonpharmaceutical treatment arms in a cost—utility model. The nonresponder and untreated health states were identical except for the statement indicating whether treatment is received. The three health states are presented in the Supplementary material.

The health states were drafted based on literature review, interviews with clinicians, and clinical trial data. First, a

literature review was conducted to ensure that the health states would be grounded in clinical research. 14-20 This review focused on identifying the key symptoms and functional impact of adult ADHD. 21-26 This literature was used to draft structured interview guides for the clinician interviews. Then, interviews were conducted with four clinicians who had experience with adult ADHD. All four had Doctor of Medicine degrees, while two also had PhD degrees. During interviews, the clinicians responded to questions regarding symptoms and impact of adult ADHD.

Pooled data from six clinical trials were examined to ensure that health states accurately described differences between treatment responders and nonresponders. The six trials were parallel double-blind placebo-controlled trials of atomoxetine for the treatment of adult ADHD. Patients were categorized as responders if they had at least a 30% symptom reduction from baseline on either the Conners' Adult ADHD Rating Scale (CAARS) total symptom score²⁷ or Adult ADHD Investigator Symptom Rating Scale (AISRS) total score,²⁸ plus a Clinical Global Impression-ADHD Severity score ≤3. The 30% symptom-reduction cutoff is commonly used as a threshold for categorizing adult ADHD treatment response across a wide range of clinical trials and outcome measures.^{29–36}

Then, endpoint individual item scores were examined on measures of symptoms and function. Language from items distinguishing between responders and nonresponders was included in the health states. For example, the health states included a statement "trouble concentrating when people are speaking to you", which roughly corresponds to item 14 of the CAARS and item 5 of the AISRS. In the trials, responders had, respectively, mean scores of 0.67 and 0.74 on these two items, compared with 1.90 and 2.02 for nonresponders. Health states also included "trouble completing tasks and getting things done on time at work and at home", which corresponds to item 22 of the Adult ADHD Quality of Life Measure (AAQOL).³⁷ In the pooled trial data, responders had a mean score of 2.22 on this item, compared with 3.50 for nonresponders.

The responder health state says symptoms occur "once in a while" with "minimal consequences". The nonresponder and untreated health states say symptoms occur "often" with "real consequences". These phrases indicating symptom severity and impact were taken directly from response options of the CAARS, AISRS, and AAQOL. The specific response options were selected based on mean scores for responders and nonresponders in the six clinical trials described above. For example, the CAARS response option of 1 ("just a little,

once in a while") was used to represent symptom severity of responders, whereas the response option of 2 ("pretty much, often") was used to represent symptom severity of nonresponders. Similarly, the AISRS response option 1 ("minimal functional consequences") was used to represent impact in the responder health state, while 2 ("real functional consequences") was used to represent impact in the nonresponder health state. On the AAQOL, response option 2 ("a little") corresponded to responders, while 4 ("often") corresponded to nonresponders. For the current health states, symptom severity of untreated patients was assumed to be similar to that of nonresponders.

Health states were drafted based on the language in these clinical trial measures while taking into account clinician input and published literature. Draft health states were tested in a pilot study in London with 26 general population participants (13 female; mean age =42.2 years, 50.0% white) recruited through a newspaper advertisement. The draft health states were administered in a TTO interview to ensure that respondents were able to understand the health states and the interview task. Participants consistently reported that the health states were clear and easy to understand. Some participants suggested minor revisions in formatting, phrasing, and explanation of the TTO task, and the health states were edited accordingly. Health states were also edited to ensure that they reflected typical patients with ADHD as closely as possible, without overestimating or underestimating the impact of this condition. For example, an early draft of the nonresponder health state specified that the patient could not maintain employment. The statement was dropped during the pilot study because: 1) the statement appeared to represent a more severe impact than would be experienced by a typical patient; 2) the severity of the statement seemed to distract respondents from attending to the key symptoms and functional impairment of ADHD; and 3) unlike the other statements, it was not specifically tied to clinical trial measures.

Participants

Participants were required to be at least 18 years old, able to understand the procedures, able to give written informed consent, and residing in the UK. Inclusion criteria did not specify any particular clinical characteristics because interviews were intended to yield utilities that may be used in submissions to agencies like the National Institute for Health and Care Excellence (NICE), most of whom prefer that utilities represent general population values.^{38–40}

Of the potential participants who responded by telephone to newspaper or online advertisements, 203 were reached for screening. Of the 203 screened participants, 199 were eligible, 174 were scheduled for interviews, and 160 participants attended and completed their interviews. Two of the 160 participants were unable to complete the utility interview procedures. Thus, a total of 158 interviews were completed.

Utility interview procedures and scoring

Interviews were conducted in private conference rooms in Edinburgh and London during October 2012. Procedures and materials were approved by an independent institutional review board, and participants provided written informed consent prior to completing study measures.

Health states were first rated using a visual analogue scale (VAS) intended to introduce participants to the content. Then, health-state utilities were obtained using the TTO method, which has previously been described in detail.⁵ TTO interviews were conducted with a 10-year time horizon, which is consistent with the influential Measurement and Valuation of Health study that elicited valuations from the general public for EQ-5D health states.⁴¹ NICE has recommended that, when EQ-5D utilities are not available, utilities for UK costutility models should be derived using methods similar to Measurement and Valuation of Health methodology.⁴² In the current study, participants were offered choices alternating between greater and lesser durations in full health, varying in 1-year increments (ie, 9 years, 1 year, 8 years, 2 years, 7 years, etc).

For each health state rated as preferable to being dead in the TTO task, the utility value was calculated based on the choice in which the respondent is indifferent between *y* years in the health state being evaluated and *x* years in full health (followed by *y*–*x* years dead). The resulting utility estimate (*u*) is calculated as:

$$u = x/y$$
.⁴³ [1]

If participants indicated that a health state was worse than dead (which seldom occurred in the current study), the interviewer altered the task so that respondents were offered a choice between immediate death (alternative 1) and a 10-year life span (alternative 2) beginning with varying amounts of time in the health state being rated, followed by full health for the remainder of the 10-year timeframe. For these negative utilities, the current study used a bounded scoring approach, which is commonly used to avoid highly skewed distributions.^{5,43,44} This scoring approach limits

the score range of health states worse than dead between 0 and -1; formula:

$$u = -x/t [2]$$

where *x* is time in full health and *t* is the total life span of alternative 2 in the TTO choice.

Statistical analysis procedures

SAS statistical software version 9.2 (SAS Institute Inc., Cary, NC, USA) was used for all analyses. Continuous variables including utilities were summarized in terms of means and SDs, and categorical demographic variables were summarized as frequencies and percentages. A series of independent *t*-tests (for continuous variables) and chi-square analyses (for categorical variables) were conducted to compare utilities and demographic characteristics of the samples from London and Edinburgh. Pairwise comparisons between health states, using *t*-tests, were conducted to examine differences in preference.

Results

Sample description

A total of 158 participants completed the interview. The sample had a mean age of 47.0 years (SD =14.4) and was 49.4% female. Ethnicity, marital status, employment status, and education level are summarized in Table 1.

Only one participant reported being diagnosed with ADHD, but 33 participants (20.9%) reported knowing a friend or family member diagnosed with ADHD. A total of 35 participants (22.2%) reported being diagnosed with a mental health condition, including alcohol abuse (n=2), substance abuse (n=1), anxiety (n=18), Asperger's syndrome (n=2), and depression (n=26).

There were significant differences between the London (n=78) and Edinburgh (n=80) samples in ethnicity and marital status (Table 1).

The Edinburgh sample had a higher percentage of white participants while the London sample had a higher percentage of single participants.

TTO utilities and VAS scores

Mean (SD) TTO utilities were 0.82 (0.17), 0.68 (0.28), and 0.67 (0.28) for health states representing adult ADHD responders (health state A), nonresponders (health state B), and untreated patients (health state C), respectively (Table 2). There were no significant differences in health-state utilities between the London and Edinburgh samples. VAS ratings followed the same pattern of relationships among the three health states: A, 80.0 (SD =16.9); B, 47.1 (SD =17.2); and C, 44.4 (SD =20.4).

In the TTO interviews, 92 participants (58.2%) rated health state A greater than B, while 97 (61.4%) rated A greater than C. The majority of participants rated B

Table I Demographic characteristics

Characteristics	London (n=78)	Edinburgh (n=80)	Total sample (n=158)	P-value ^a
Sex, n (%)				
Male	41 (52.6)	39 (48.8)	80 (50.6)	0.63
Female	37 (47.4)	41 (51.3)	78 (49.4)	
Ethnicity, n (%)				
White	55 (70.5)	77 (96.3)	132 (83.5)	0.0002
Black	7 (9.0)	0 (0.0)	7 (4.4)	
Asian	8 (10.3)	I (I.3)	9 (5.7)	
Mixed	8 (10.3)	2 (2.5)	10 (6.3)	
Marital status, n (%)				
Single	42 (53.8)	22 (27.5)	64 (40.5)	0.0033
Married	24 (30.8)	37 (46.3)	61 (38.6)	
Other ^b	12 (15.4)	21 (26.3)	33 (20.9)	
Employment status, n (%)				
Work full time or part time	49 (62.8)	43 (53.8)	92 (58.2)	0.25
Other ^c	29 (37.2)	37 (46.3)	66 (41.8)	
Education level, n (%)				
Completed university degree	32 (41.0)	28 (35.0)	60 (38.0)	0.44
Did not complete university degree	46 (59.0)	52 (65.0)	98 (62.0)	

Notes: *P-values are for comparisons between the London and Edinburgh samples. Comparisons were performed with independent Student's t-tests for continuous variables and chi-square analyses for categorical variables. bincludes divorced, separated, and widowed. clncludes homemaker/housewife, student, unemployed, retired, and disabled.

Table 2 Time trade-off utility scores

Health states describing adult ADHD Mean (standard deviation)	London (n=78)	Edinburgh (n=80)	Total sample (n=158)	P-value ^a
(B) Treatment nonresponder	0.67 (0.29)	0.69 (0.28)	0.68 (0.28)	0.61
(C) Untreated	0.65 (0.30)	0.68 (0.27)	0.67 (0.28)	0.50

Note: ^aP-values are from t-tests comparing the London and Edinburgh samples. **Abbreviation:** ADHD, attention-deficit/hyperactivity disorder.

and C as equal (n=125; 79.1%), although some rated B greater than C (n=23; 14.6%) and others rated C greater than B (n=10; 6.3%). t-tests comparing mean TTO utilities found that A was significantly greater than B (t=10.0; t=0.001) and C (t=10.2; t=0.001). The small difference between health states B and C was also statistically significant (t=2.1; t=0.037).

The great majority of the 158 respondents provided positive utility scores for all health states, indicating that the three health states were almost always perceived as preferable to being dead. Negative utility scores, which indicate that a health state was perceived as worse than being dead, were provided by one respondent for health state A, three respondents for health state B, and three respondents for health state C.

Discussion

Results reflected clear utility differences between adult ADHD health states representing treatment responders and health states representing nonresponders or untreated patients. Utility values identified in the current study were in a similar range to TTO utilities previously reported by Lloyd et al⁷ for childhood ADHD. The current utility of 0.82 for a health state representing treatment responders who have symptoms "once in a while" with "minimal consequences" is between the values Lloyd et al reported for childhood ADHD with severity levels of "normal" (0.84) and "borderline to mildly ill" (0.79).⁷ The current scores of 0.68 and 0.67 for health states describing nonresponders and untreated patients, respectively, are between the previously reported values for childhood ADHD with severity levels of "borderline to mildly ill" (0.79) and "moderately to markedly ill" (0.58).

While the vignette-based utility assessment method used in the current study is well-suited for identifying utilities associated with specific medical or psychiatric conditions, the limitations of this method should be acknowledged. For example, it is difficult to know the extent to which vignette-based utility values are comparable to utilities derived from other approaches, such as generic preference-based measures. In addition, the usefulness of resulting utility scores is limited

by the content of the health states. For example, the current health states only reflected two levels of symptom severity (the nonresponder and untreated health states had the same descriptions of symptom severity). ADHD is a heterogeneous condition with a wide range of severity levels that could be represented in additional health states. In addition, patients' health-related quality of life is likely to be influenced by other factors that were not included in these health states, such as treatment-related adverse events. Therefore, the current study should be considered an early step in estimation of utilities associated with adult ADHD, and future research is needed to identify utilities of a wider range of health states representing more severity levels and specific treatment attributes.

The supporting information used when drafting health states could also have an impact on results. Although health states were drafted based on comprehensive supporting information drawn from literature review, clinical trial results, and clinician interviews, no patients with adult ADHD were directly interviewed. However, the clinical trial results included a patient-reported measure and a clinician-rated measure of all *Diagnostic and Statistical Manual of Mental Disorders, Fourth Edition* ADHD criteria completed based on interviews with patients. Therefore, while it is possible that patients would report different symptoms than those reflected in the current health states, these health states should largely reflect the patient experience.

It should also be noted that the clinical trial results used to support the health states were from studies focused on a single pharmaceutical treatment (ie, atomoxetine). In these trials, the symptoms that were found to differ between responders and nonresponders covered a broad range of ADHD criteria, so it is reasonable to expect that these differences are representative of improvement that may be experienced with other ADHD treatments. Furthermore, respondents were not oriented to any particular treatment, and the responder health state did not specify any particular type of treatment ("You receive treatment for ADHD. This may include medication [tablets every day] and/or psychological treatment"). Still, it is possible that different treatments could have different effects on ADHD symptoms and,

thus, the improved symptoms described in the responder health state may not be equally representative of patients responding to pharmaceutical treatments other than atomoxetine. Fortunately, the use of published literature and clinician interviews as additional sources of health-state support help to ameliorate this concern.

Future studies may also examine adult ADHD utilities with different methodology. For example, NICE states a preference for utilities based on the EQ-5D in order to maximize consistency across appraisals. However, the guidelines indicate that there is flexibility in situations when the EQ-5D may not apply to a particular population or medical condition. With regard to the current study, the EO-5D was not considered optimal for evaluation of ADHD health states because none of the five items assess attention or other ADHD symptoms. Therefore, the current study followed methods that NICE has recommended as an alternative: "direct valuations of descriptions of health states" with "the time trade-off method in a representative sample of the UK population... to retain methodological consistency with the methods used to value the EQ-5D". 42 Future research may determine the extent to which alternative methods, such as the EQ-5D or other variations of TTO interviews, would yield different utilities. One possible approach would be to add an ADHD-specific item to the EQ-5D. This approach, often called a "bolt-on", has recently been examined as a way to make the generic EQ-5D applicable to a wider range of specific medical conditions.⁴⁵

Limitations associated with sample characteristics should also be acknowledged. The current study was conducted in two UK locations with a broad sample of respondents, without taking the respondents' health status into account. However, the sample was not recruited to be representative of the UK general population, and it is not known whether utilities may be different in a nationally representative sample. Still, despite demographic differences between the London and Edinburgh samples, there were no significant differences in utility between the two geographic locations, which suggests that health-state preferences may not vary greatly among UK samples.

A possible limitation of the current health states is that they named ADHD as the disease condition. Previous research has yielded mixed results regarding the impact of including the condition label in health states. Some research has suggested that including the condition label can affect utility values, while others have reported situations where the label did not affect values. 46,47 The impact of the label may depend on the specific medical condition and its severity. For example, one recent study found that a cancer label

resulted in lower utilities, while the label of irritable bowel syndrome had no significant effect on utility. 48 In light of these mixed results, recommendations vary and it is common to include the condition label in health state descriptions. Although some researchers recommend excluding the label to avoid potential impact on utility values, it can also be argued that the label leads to more accurate representation of the health state because patients who live in a health state experience it with knowledge of the label. In the current study, it was decided to include the ADHD label to ensure that participants viewed all health states as true health conditions requiring treatment, despite the mild symptoms of the responder health state. Future research may examine the extent to which the ADHD label influences health utility values.

Another aspect of the study design that could influence results is that utilities were derived from preferences of general population respondents, instead of patients with adult ADHD. As frequently discussed in previous literature, both the general population and patient approaches have advantages and disadvantages, and there may be differences in utilities from the two types of samples. 5,49-51 The generalpopulation approach is consistent with the most frequently cited methodological guidance provided by health technology assessment agencies.^{38–40,52} Theoretically, general population values can be considered representative of a societal view that may be relevant for decision-making regarding public health care funding. On the other hand, patients with personal experience may have greater insight into the content of health-state descriptions and, therefore, patient utility valuations may reflect a more accurate understanding of the health states. It is not known whether patient-based utilities would differ from the utility scores reported in the current study.

Conclusion

Despite potential limitations, the current study provides utilities that may be useful for calculating quality-adjusted life years in cost—utility models of treatment for adult ADHD. It is hoped that future research can build on this study by providing a more detailed set of utilities, thus leading to more accurate modeling to demonstrate the value of various treatments for adult ADHD.

Acknowledgments

Funding for this study was provided by Eli Lilly and Company. The authors would like to thank: Kristina Boye for scientific consulting; Christine Thompson for statistical programming; Amanda Landrian, Mary Kay Margolis, and

Anna Tate for assistance with data collection; and Aria Gray and Amara Tiebout for production assistance.

Disclosure

Two of the authors (Virginia Sutton Haynes, Foula Televantou) are employees of Lilly, but their input into the conceptualization and interpretation of this study represented their own opinions, rather than those of the company. All other authors are employees of Evidera, a company that received funding from Lilly for this research. The authors report no other conflicts of interest in this work.

References

- Biederman J, Monuteaux MC, Mick E, et al. Young adult outcome of attention deficit hyperactivity disorder: a controlled 10-year follow-up study. *Psychol Med.* 2006;36(2):167–179.
- Faraone SV, Biederman J. What is the prevalence of adult ADHD? Results of a population screen of 966 adults. *J Atten Disord*. 2005;9(2): 384–391.
- Kessler RC, Adler LA, Barkley R, et al. Patterns and predictors of attention-deficit/hyperactivity disorder persistence into adulthood: results from the national comorbidity survey replication. *Biol Psychiatry*. 2005;57(11):1442–1451.
- Mannuzza S, Klein RG, Moulton JL 3rd. Persistence of Attention-Deficit/Hyperactivity Disorder into adulthood: what have we learned from the prospective follow-up studies? *J Atten Disord*. 2003;7(2): 93–100.
- Brazier JR, Ratcliffe J, Salomon JA, Tsuchiya A. Measuring and Valuing Health benefits for Economic Evaluation. New York: Oxford University Press; 2007.
- Coghill D, Spende Q, Barton J, et al. Measuring Quality of Life in Children with Attention-Deficit-Hyperactivity Disorder in the UK. Poster presented at 16th World Congress of the International Association for Child and Adolescent Psychiatry and Allied Professions (IACAPAP); 22–26 August 2004; Berlin, Germany. 2004.
- Lloyd A, Hodgkins P, Sasane R, et al. Estimation of utilities in attentiondeficit hyperactivity disorder for economic evaluations. *Patient*. 2011; 4(4):247–257.
- Matza LS, Secnik K, Rentz AM, et al. Assessment of health state utilities for attention-deficit/hyperactivity disorder in children using parent proxy report. *Qual Life Res.* 2005;14(3):735–747.
- Secnik K, Matza LS, Cottrell S, Edgell E, Tilden D, Mannix S. Health state utilities for childhood attention-deficit/hyperactivity disorder based on parent preferences in the United kingdom. *Med Decis Making*. 2005;25(1):56–70.
- Van Brunt K, Matza LS, Classi PM, Johnston JA. Preferences related to attention-deficit/hyperactivity disorder and its treatment. *Patient Prefer Adherence*. 2011;5:33–43.
- Zimovetz EA, Setyawan J, Beard SM, Hodgkins P. Systemic review of health-state utilities in Attention Deficit Hyperactivity Disorder (ADHD). Presented at: ISPOR 15th Annual European Congress; November 3–7, 2012; Berlin.
- Laing A, Aristides M. Attention deficit hyperactivity disorder (ADHD) in adults: SF-6D utilities from SF-36 scores in a randomised trial. Presented at: 8th Annual European Congress of the International Society for Pharmacoeconomics and Outcomes Research; November 6–8; 2005: Florence.
- Brazier J, Roberts J, Deverill M. The estimation of a preferencebased measure of health from the SF-36. *J Health Econ.* 2002;21(2): 271–292.
- Aapro M, Abrahamsson PA, Body JJ, et al. Guidance on the use of bisphosphonates in solid tumours: recommendations of an international expert panel. *Ann Oncol.* 2008;19(3):420–432.

 Body JJ, Diel IJ, Bell R, et al. Oral ibandronate improves bone pain and preserves quality of life in patients with skeletal metastases due to breast cancer. *Pain.* 2004;111(3):306–312.

- Coleman RE. Metastatic bone disease: Clinical features, pathophysiology and treatment strategies. Cancer Treat Rev. 2001;27(3):165–176.
- Coleman RE, Guise TA, Lipton A, et al. Advancing treatment for metastatic bone cancer: consensus recommendations from the Second Cambridge Conference. Clin Cancer Res. 2008;14(20): 6387–6395.
- Rosen LS, Gordon D, Tchekmedyian NS, et al. Long-term efficacy and safety of zoledronic acid in the treatment of skeletal metastases in patients with nonsmall cell lung carcinoma and other solid tumors: a randomized, Phase III, double-blind, placebo-controlled trial. *Cancer*. 2004;100(12):2613–2621.
- Smith HS. Painful osseous metastases. Pain Physician. 2011;14(4): E373–E403.
- Weinfurt KP, Li Y, Castel LD, et al. The significance of skeletal-related events for the health-related quality of life of patients with metastatic prostate cancer. *Ann Oncol.* 2005;16(4):579–584.
- Adler L, Cohen J. Diagnosis and evaluation of adults with attentiondeficit/hyperactivity disorder. *Psychiatr Clin North Am.* 2004;27(2): 187–201.
- Biederman J, Faraone SV, Spencer TJ, Mick E, Monuteaux MC, Aleardi M. Functional impairments in adults with self-reports of diagnosed ADHD: A controlled study of 1001 adults in the community. *J Clin Psychiatry*. 2006;67(4):524–540.
- Kooij SJ, Bejerot S, Blackwell A, et al. European consensus statement on diagnosis and treatment of adult ADHD: The European Network Adult ADHD. BMC Psychiatry. 2010;10:67.
- Mannuzza S, Castellanos FX, Roizen ER, Hutchison JA, Lashua EC, Klein RG. Impact of the impairment criterion in the diagnosis of adult ADHD: 33-year follow-up study of boys with ADHD. *J Atten Disord*. 2011;15(2):122–129.
- Riccio CA, Wolfe M, Davis B, Romine C, George C, Lee D. Attention Deficit Hyperactivity Disorder: manifestation in adulthood. *Arch Clin Neuropsychol.* 2005;20(2):249–269.
- Wilens TE, Biederman J, Faraone SV, Martelon M, Westerberg D, Spencer TJ. Presenting ADHD symptoms, subtypes, and comorbid disorders in clinically referred adults with ADHD. *J Clin Psychiatry*. 2009;70(11):1557–1562.
- Conners CK, Erhardt D, Sparrow E. Conners' Adult ADHD Rating Scales, Technical Manual. New York, NY: Multi-Health Systems Inc; 1999.
- Adler LA, Spencer T, Faraone SV, et al. Validity of pilot adult ADHD self-report scale (pilot ASRS) to rate adult ADHD symptoms. Presented at: American Academy of Child and Adolescent Psychiatry; October 14–19 2003; Miami, FL.
- Adler LA, Zimmerman B, Starr HL, et al. Efficacy and safety of OROS methylphenidate in adults with attention-deficit/hyperactivity disorder: a randomized, placebo-controlled, double-blind, parallel group, dose-escalation study. J Clin Psychopharmacol. 2009;29(3): 239–247.
- Biederman J, Mick E, Surman C, et al. A randomized, placebocontrolled trial of OROS methylphenidate in adults with attentiondeficit/hyperactivity disorder. *Biol Psychiatry*. 2006;59(9):829–835.
- Biederman J, Mick E, Surman C, et al. A randomized, 3-phase, 34-week, double-blind, long-term efficacy study of osmotic-release oral systemmethylphenidate in adults with attention-deficit/hyperactivity disorder. *J Clin Psychopharmacol*. 2010;30(5):549–553.
- Casas M, Rösler M, Sandra Kooij JJ, et al. Efficacy and safety of prolonged-release OROS methylphenidate in adults with attentiondeficit/hyperactivity disorder: a 13-week, randomized, double-blind, placebo-controlled, fixed-dose study. World J Biol Psychiatry. 2013;14(4):268–281.
- Medori R, Ramos-Quiroga JA, Casas M, et al. A randomized, placebocontrolled trial of three fixed dosages of prolonged-release OROS methylphenidate in adults with attention-deficit/hyperactivity disorder. *Biol Psychiatry*. 2008;63(10):981–989.

- 34. Retz W, Rosler M, Ose C, et al. Multiscale assessment of treatment efficacy in adults with ADHD: a randomized placebo-controlled, multi-centre study with extended-release methylphenidate. World J Biol Psychiatry. 2012;13(1):48-59.
- 35. Weisler RH, Pandina GJ, Daly EJ, Cooper K, Gassmann-Mayer C; Investigators ATTS. Randomized clinical study of a histamine H3 receptor antagonist for the treatment of adults with attention-deficit/ hyperactivity disorder. CNS Drugs. 2012;26(5):421-434.
- 36. Winhusen TM, Somoza EC, Brigham GS, et al. Does treatment of Attention Deficit Hyperactivity Disorder (ADHD) enhance response to smoking cessation intervention in ADHD smokers? A randomized trial. J Clin Psychiatry. 2010;71(12):1680-1688.
- 37. Brod M, Johnston J, Able S, Swindle R. Validation of the adult attentiondeficit/hyperactivity disorder quality-of-life Scale (AAQoL): a diseasespecific quality-of-life measure. Qual Life Res. 2006;15(1):117-129.
- 38. Brazier J. Valuing health States for use in cost-effectiveness analysis. Pharmacoeconomics. 2008;26(9):769-779.
- 39. NICE (National Institute for Health and Clinical Excellence). Process and methods guides: Guide to the methods of technology appraisal 2013. London: National Institute for Health and Care Excellence; 2013. Available from: http://www.nice.org.uk/media/ D45/1E/GuideToMethodsTechnologyAppraisal2013.pdf. Accessed September 10, 2013.
- 40. PBAC. Guidelines for preparing submissions to the Pharmaceutical Benefits Advisory Committee. Canberra, Australia: PBAC; June 2013; Version 4.4: http://www.pbac.pbs.gov.au/content/information/ printable-files/pbacg-book.pdf. Accessed May 22, 2014.
- 41. Dolan P. Modeling valuations for EuroQol health states. Med Care. 1997;35(11):1095-1108.
- 42. NICE (National Institute for Health and Clinical Excellence). Guide to the methods of technology appraisal. London: National Institute for Health and Care Excellence; 2008. Available from: http://www.nice. org.uk/media/b52/a7/tamethodsguideupdatedjune2008.pdf. Accessed September 10, 2013.

- 43. Rowen D, Brazier J. Health Utility Measurement. In: Glied S, Smith PC, editors. The Oxford Handbook of Health Economics. New York: Oxford University Press; 2011:788-813.
- 44. Dolan P, Gudex C, Kind P, Williams A. The time trade-off method: results from a general population study. Health Econ. 1996;5(2):
- 45. Yang Y, Brazier J, Tsuchiya A. Effect of adding a sleep dimension to the EQ-5D descriptive system: a "bolt-on" experiment. Med Decis Making. 2014;34(1):42-53.
- 46. Gerard K, Dobson M, Hall J. Framing and labelling effects in health descriptions: quality adjusted life years for treatment of breast cancer. J Clin Epidemiol. 1993;46(1):77-84.
- 47. Sackett DL, Torrance GW. The utility of different health states as perceived by the general public. J Chronic Dis. 1978;31(11):697–704.
- 48. Rowen D, Brazier J, Tsuchiya A, Young T, Ibbotson R. It's all in the name, or is it? The impact of labeling on health state values. Med Decis Making. 2012;32(1):31-40.
- 49. Krabbe PF, Tromp N, Ruers TJ, van Riel PL. Are patients' judgments of health status really different from the general population? Health Qual Life Outcomes. 2011;9:31.
- 50. Peeters Y, Stiggelbout AM. Health state valuations of patients and the general public analytically compared: a meta-analytical comparison of patient and population health state utilities. Value Health. 2010:13(2):306-309.
- 51. Ubel PA, Loewenstein G, Jepson C. Whose quality of life? A commentary exploring discrepancies between health state evaluations of patients and the general public. Qual Life Res. 2003;12(6):599-607.
- 52. CADTH (Canadian Agency for Drugs and Technologies in Health). Guidelines for the economic evaluation of health technologies: Canada. 3rd ed. Ottawa, ON: Canadian Agency for Drugs and Technologies in Health: 2006.

Supplementary materials

Table SI Health state A: treatment responder

ADHD

- A doctor has diagnosed you with attention-deficit/hyperactivity disorder, which is sometimes called ADD or ADHD.
- o You first had these problems as a young child, and the problems have persisted into adulthood.

Treatments

o You receive treatment for ADHD. This may include medication (tablets every day) and/or psychological treatment.

• Symptoms

- ONCE IN A WHILE, you have the symptoms below:
- o You have a little difficulty concentrating and keeping your attention focused.
- You have problems organizing tasks and activities.
- O You procrastinate and have a little difficulty getting started on tasks.
- o You have a little difficulty finding important things like your keys or wallet.
- o You feel a little internal restlessness and tension, and you appear a little restless to others even when you are sitting. You fidget and have a little trouble sitting still.

• Functioning

- Your ADHD has some <u>MINIMAL CONSEQUENCES</u> for your life.
- o Your ADHD symptoms have a minimal impact on your social interaction and relationships.
 - Once in a while, you have a little trouble concentrating when people are speaking to you.
 - If you are in a relationship with a spouse or significant other, your ADHD symptoms cause minimal difficulties, such as tension and arguments once in a while.
- Your ADHD symptoms have a minimal impact on your productivity.
 - Once in a while, you have a little trouble completing tasks and getting things done on time at work and at home.
 - Once in a while, you have a little difficulty remembering appointments.

Table S2 Health state B: treatment nonresponder

ADHD

- A doctor has diagnosed you with attention-deficit/hyperactivity disorder, which is sometimes called ADD or ADHD.
- O You first had these problems as a young child, and the problems have persisted into adulthood.

Treatments

You receive treatment for ADHD. This may include medication (tablets every day) and/or psychological treatment.

Symptoms

- o You OFTEN have the symptoms below:
- You have difficulty concentrating and keeping your attention focused.
- o You have problems organizing tasks and activities.
- You procrastinate and have difficulty getting started on tasks.
- O You have difficulty finding important things like your keys or wallet.
- o You feel internal restlessness and tension, and you appear restless to others even when you are sitting. You fidget and have trouble sitting still.

Functioning

- o Your ADHD has **REAL CONSEQUENCES** for your life.
- o Your ADHD symptoms have an impact on your social interaction and relationships.
 - You often have trouble concentrating when people are speaking to you.
 - If you are in a relationship with a spouse or significant other, your
 ADHD symptoms often cause difficulties, such as tension and arguments.
- Your ADHD symptoms have an impact on your productivity.
 - You often have trouble completing tasks and getting things done on time at work and at home.
 - You often have difficulty remembering appointments.

Matza et al Dovepress

Table S3 Health state C: untreated

ADHD

- o A doctor has diagnosed you with attention-deficit/hyperactivity disorder, which is sometimes called ADD or ADHD.
- You first had these problems as a young child, and the problems have persisted into adulthood.

Treatments

You do not receive treatment for ADHD.

Symptoms

- O You OFTEN have the symptoms below:
- o You have difficulty concentrating and keeping your attention focused.
- You have problems organizing tasks and activities.
- You procrastinate and have difficulty getting started on tasks.
- o You have difficulty finding important things like your keys or wallet.
- o You feel internal restlessness and tension, and you appear restless to others even when you are sitting. You fidget and have trouble sitting still.

Functioning

- o Your ADHD has REAL CONSEQUENCES for your life.
- Your ADHD symptoms have an impact on your social interaction and relationships.
 - You often have trouble concentrating when people are speaking to you.
 - If you are in a relationship with a spouse or significant other, your ADHD symptoms often cause difficulties, such as tension and arguments.
- Your ADHD symptoms have an impact on your productivity.
 - You often have trouble completing tasks and getting things done on time at work and at home.
 - You often have difficulty remembering appointments.

Patient Preference and Adherence

Publish your work in this journal

Patient Preference and Adherence is an international, peer-reviewed, open access journal that focuses on the growing importance of patient preference and adherence throughout the therapeutic continuum. Patient satisfaction, acceptability, quality of life, compliance, persistence and their role in developing new therapeutic modalities and compounds to optimize

clinical outcomes for existing disease states are major areas of interest for the journal. This journal has been accepted for indexing on PubMed Central. The manuscript management system is completely online and includes a very quick and fair peer-review system, which is all easy to use. Visit http://www.dovepress.com/testimonials.php to read real quotes from published authors.

 $\textbf{Submit your manuscript here:} \ \texttt{http://www.dovepress.com/patient-preference-and-adherence-journal} \\$

